Author's response to reviews

Title: Women with disability: the experience of maternity care during pregnancy, labour and birth and the postnatal period

Authors:

Maggie Redshaw (maggie.redshaw@npeu.ox.ac.uk)
Reem Malouf (reem.malouf@npeu.ox.ac.uk)
Haiyan Gao (haiyan.gao@cts.u.ox.ac.uk)
Ron Gray (ron.gray@npeu.ox.ac.uk)

Version: 2 Date: 14 May 2013

Author's response to reviews:

1) The formatting has been changed to remove the numbers.
2) Access to the data were granted by the CQQ and this point has been added to the acknowledgments section of the paper.
3) A competing interests section has been added.
4) An authors' contributions section has been added.